Author's response to reviews

Title: Enhanced Relapse Prevention for bipolar disorder - ERP trial. A cluster randomised controlled trial to assess the feasibility of training care coordinators to offer enhanced relapse prevention for bipolar disorder.

Authors:

Fiona Lobban (fiona.lobban@manchester.ac.uk)
Carol Gamble (c.gamble@liverpool.ac.uk)
Peter Kinderman (p.kinderman@liverpool.ac.uk)
Lee Taylor (lee.taylor@penninecare.nhs.uk)
Claire Chandler (claire.chandler@liverpool.ac.uk)
Elizabeth Tyler (e.tyler@liverpool.ac.uk)
Sarah Peters (s.peters@liverpool.ac.uk)
Eleanor Pontin (e.pontin@liverpool.ac.uk)
William Sellwood (sellwood@liverpool.ac.uk)
Richard K Morriss (richard.morriss@nottingham.ac.uk)

Version: 3 Date: 1 February 2007

Author's response to reviews:

Thank you for your email outlining formatting changes required. I have addressed each of these and ensured any changes are also made on the submission pages.

Kind regards
Fiona